Author's response to reviews

Title: Mesenchymal and stemness circulating tumor cells in early breast cancer diagnosis

Authors:

  guislaine barriere (guislainebarriere@yahoo.fr)
  alain riouallon (riouallon@clinique-colombier.fr)
  joel renaudie (renjo@clinique-colombier.fr)
  michel tartary (michel.tartary@numericable.com)
  Prof rigaud michel (rigaud.michel@yahoo.fr)

Version: 5  Date: 17 February 2012

Author's response to reviews:

As Sabine Kasimir-Bauer asked, we have corrected spelling errors (20) and the manuscript was effectively read again by a native English scientist.

As asked by Lisa Ryden, we add a sentence to address the issue of an ongoing trial in our lab demonstrating that 8% of patients upon 400 are positive for CTC by using the basic AdnaGen test.